Author's response to reviews

Title: The respiratory health of urban Indigenous children aged less than 5 years: study protocol for a prospective cohort study

Authors:

Kerry K Hall (kk.hall@qut.edu.au)
Anne B Chang (anne.chang@menzies.edu.au)
Theo P Sloots (t.sloots1@uq.edu.au)
Jennie Anderson (jennie@murrihealthgroup.org.au)
Anita Kemp (anita@murrihealthgroup.org.au)
Janet Hammill (janet.hammill@uq.edu.au)
Michael Otim (michael.otim@acu.edu.au)
Kerry-Ann F O'Grady (kerryann.ogrady@qut.edu.au)

Version: 2 Date: 10 February 2015

Author's response to reviews:

please find listed below the requested changes and clarification of information, copies of relevant funding and ethics documents have been forwarded to BMCSeriesEditorial@biomedcentral.com as requested.

1. Line Numbering:
Line numbering has been added as requested.

2. Trial Registration Number:
Trial registration and date of trial registration have been amended per instruction

3. Study Protocol Proofs:

a) Ethical and Funding Approval Documentation
Ethics has been provided by Queensland Children’s Health Services

b) Funding
Funding is from a QLD Government Smart Futures Fellowship, University of Queensland Foundation Research Excellence Award, Queensland Children’s Health Foundation Project Grant and a NHMRC Career Development Fellowship awarded to Dr O’Grady

c) Study status
Study status has been amended to read ongoing study

d) Related Articles There are currently no published articles relating to this study